Author's response to reviews

Title: A person presenting with common yew intoxication: a case presentation.

Authors:

Martin Valis (valismar@seznam.cz)
Jaromír Kočí (jaromir.koci@fnhk.cz)
David Tuček (tucekdav@fnhk.cz)
Tomáš Lutonský (tomas.lutonsky@fnhk.cz)
Jana Kopová (jana.kopova@fnhk.cz)
Petr Bartoň (petr.barton.jaromer@seznam.cz)
Oldřich Vyšata (vysatao@gmail.com)
Dagmar Krajířová (dagmar.krajickova@fnhk.cz)
Jan Korábečný (jan.korabecny@fnhk.cz)
Jiří Masopust (masopjir@seznam.cz)
Ludovít Klzo (ludek.klzo@seznam.cz)

Version: 2 Date: 17 September 2013

Author's response to reviews:

In Hradec Kralove 17.09.2013

Dear Editor,

First we want to thank both reviewers and editor for their valuable ideas how to improve our manuscript. We have made all corrections according to reviewers’ and editor suggestions wherever possible. Only the results of laboratory tests and blood gas were not added because they had no influence on the course of disease.

Yours sincerely,

Martin Vališ
Department of Neurology, Charles University and University Hospital
Sokolská 581
500 05 Hradec Králové
Czech Republic